Author’s response to reviews

Title: Development of research integrity in France is on the rise: the introduction of research integrity officers was a progress

Authors:

Herve Maisonneuve (hervemaisonneuve@gmail.com)

Version: 2 Date: 27 Aug 2019

Author’s response to reviews:

Thanks for the useful comments

Corrections were done

I kept the note on the WCRI poster for transparency

MERCI